Author's response to reviews

Title: Congenital cystic eye associated with a low-grade cerebellar lesion that spontaneously regressed

Authors:

Maria Giuseppina Cefalo (mpiacefalo@gmail.com)
Giovanna Stefania Colafati (gstefania.colafati@opbg.net)
Antonino Romanzo (antonino.romanzo@opbg.net)
Alessandra Modugno (ocularistica@tin.it)
Rita De Vito (rita.devito@opbg.net)
Angela Mastronuzzi (angela.mastronuzzi@opbg.net)

Version: 4 Date: 29 May 2014

Author's response to reviews: see over
Dear Editor, we are grateful that you accept the manuscript for publication in BMC Ophthalmology. We revised the manuscript according to the suggestions and we highlighted the corrections in yellow in the text.

1. We note that on page 8 you state that informed consent was obtained from the patient for the publication of this case report. However, on page 4 you state that the case is 6 months old. Can you please confirm that parental consent was obtained for the publication of this case report, and amend the consent section as appropriate.

   *Consent section has been revised.*

2. Please change the 'declaration of conflicting interests' heading to: 'competing interests'.

   *We revised according to suggestion.*

3. Please carefully copyedit the manuscript, correcting for typographical/grammatical errors (e.g. please correct the typos in the 'authors' contributions' section).

   *The manuscript has been copyedit.*

Best regards

Angela Mastronuzzi